Author’s response to reviews

Title: Limiting the Caesarean Section rate in low risk pregnancies is key to lowering the trend of increased abdominal deliveries: an observational study

Authors:

Ilse Delbaere (ilse.delbaere@ugent.be)
Hendrik Cammu (hendrik.cammu@uzbrussel.be)
Evelyne Martens (evelyne.martens@kindengezin.be)
Inge Tency (inge.tency@ugent.be)
Guy Martens (guy.martens@kindengezin.be)
Marleen Temmerman (marleen.temmerman@ugent.be)

Version: 4 Date: 1 December 2011

Author’s response to reviews:

Dear editor,

Please find the uploaded revised version of our manuscript.

As requested, we included a Discussion-section before our Conclusions ('Comment' was changed in 'Discussion', see first minor remark) and we cited Figure 2 within the manuscript text.

Within authors’ contributions, we used initials instead of full names and formatted this text as a paragraph. Vertical lines were moved from the tables.

Figures were checked and cropped as small as possible.

Kind regards,
Ilse Delbaere